Author's response to reviews

**Title:** Fetus in Fetu: A case report

**Authors:**

Nisreen M Khalifa (nisreen_khalifa@hotmail.com)  
Doaa W Maximous (doaadavid@hotmail.com)  
Alaa A Abd-Elsayed (alaaawny@hotmail.com)

**Version:** 3  **Date:** 1 October 2007

**Author's response to reviews:**

Title is changed to Fetus in fetu: A case report.  
We renamed the "Background" section to "Introduction".  
We included a "Consent" section  
We included a subheading for the "Competing Interests" statement.